Author's response to reviews

Title: Substance Abuse and Psychiatric Co-morbidity as Predictors of Premature Mortality in Swedish Drug Abusers: A Prospective Longitudinal Study 1970-2006

Authors:

Anna Nyhlén (anna.nyhlen@skane.se)
Mats Fridell (mats.fridell@psychology.lu.se)
Martin Bäckström (martin.backstrom@psychology.lu.se)
Morten Hesse (mh@crf.au.dk)
Peter Krantz (rmlu@rmv.se)

Version: 10 Date: 29 July 2011

Author’s response to reviews:

MS: 1319869118455803

Substance Abuse and Psychiatric Co-morbidity as Predictors of Premature Mortality in Swedish Drug Abusers:

A Prospective Longitudinal Study 1970-2006

Anna Nyhlén, Mats Fridell, Martin Bäckström, Morten Hesse and Peter Krantz.

18 July 2011

Dear editor,

Thank you for your letter. We have made a small correction in the abstract. The remaining manuscript is the same as the previous version.

We are sorry that our previous response from July 18 did not reach you. We submit a new version today.

With kind regards

Anna Nyhlén, MD, PhD Mats Fridell, Professor in clin psychology

Tel: +46 (46) 174995
e-mail: mats.fridell@psychology.lu.se